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OperationsOperations
29 October29 October

Handover of CS1 operations to ObservatoryHandover of CS1 operations to Observatory
DwingelooDwingeloo Operations RoomOperations Room

–– GeertGeert KuperKuper, , JurJur SlumanSluman, Yuan Tang, Yuan Tang
–– (+31) 521 595 777(+31) 521 595 777
–– observer@astron.nlobserver@astron.nl

Scheduling coordinator Scheduling coordinator MichielMichiel BrentjensBrentjens (781)(781)
Other issues & requests Hanno Other issues & requests Hanno HoltiesHolties (783)(783)

SAS/MAC interfaces used for observationsSAS/MAC interfaces used for observations
Not for 24 Not for 24 microstationmicrostation modemode
Some issues when using test environment in parallelSome issues when using test environment in parallel
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OperationsOperations room room DwingelooDwingeloo

OperationsOperations roomroom
PermanentelyPermanentely occupiedoccupied
(office (office hourshours))
PreparationPreparation + M&C of + M&C of 
observationsobservations
Data Data inspectioninspection & & 
administrationadministration
System status & System status & 
cocoöördinationrdination activitiesactivities CS1CS1
PreviousPrevious OortOort roomroom
WSRT WSRT operationsoperations as as wellwell
777; 777; observer@astron.nlobserver@astron.nl

Walk in at Walk in at anyany time!time!
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StationsStations
LBALBA

Beam validation experiments with rotated dipolesBeam validation experiments with rotated dipoles

HBAHBA
Switch problem being worked onSwitch problem being worked on

–– Boards being testedBoards being tested
–– Start assembling one tile (16 boards)Start assembling one tile (16 boards)

CabinetsCabinets
RSP III boards firmware update this afternoonRSP III boards firmware update this afternoon

–– Solves cross correlation problem found in Solves cross correlation problem found in EffelsbergEffelsberg
Mouse entries (hopefully) closedMouse entries (hopefully) closed
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NetworkNetwork
No newsNo news

Central ProcessingCentral Processing
Input ClusterInput Cluster

Work ongoing to remove from test environmentWork ongoing to remove from test environment
–– Still some problems when test/production used in parallelStill some problems when test/production used in parallel

BG/LBG/L
Short outage last Monday (no observations affected)Short outage last Monday (no observations affected)

StorageStorage
SAS/MAC operation starts with sequence no 5000SAS/MAC operation starts with sequence no 5000
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SoftwareSoftware
Step 2 development closedStep 2 development closed

Integration SAS/MAC with Stations & OLAPIntegration SAS/MAC with Stations & OLAP
SAS/MAC/OLAP now operated by ObservatorySAS/MAC/OLAP now operated by Observatory

–– Support for instruction & trouble shootingSupport for instruction & trouble shooting
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Start up operations by ObservatoryStart up operations by Observatory
Separate test Separate test –– production environmentproduction environment

Observation planning meeting on Observation planning meeting on Thursday(sThursday(s))
Coordinator Coordinator MichielMichiel BrentjensBrentjens

Install upgraded FE units in (single) HBA tileInstall upgraded FE units in (single) HBA tile
Hopefully in ~weekHopefully in ~week

Replace remaining RSP II boardsReplace remaining RSP II boards


